Author's response to reviews

Title: Minimal invasive perventricular device closure of ventricular septal defect in infants under transthoracic echocardiographic guidance: feasibility and comparison with transesophageal echocardiography

Authors:

gui-can zhang (scottie98345@sohu.com)
qiang chen (chengjianq2228@hotmail.com)
hua cao (caohua69@hotmail.com)
liang-wan chen (chenliangwan@tom.com)

Version: 2 Date: 1 March 2013

Author's response to reviews:

Dear Editor and reviewers:

We are very pleased to receive your decision letter on our MS. We studied the reviewer comments thoughtfully and found that all these comments are very constructive for revising the manuscript. We have revised the manuscript according to the comments and send it to you. My response is as follow:

We must say sorry for our poor English. We revised our MS with the help of some native.

We corrected “TTE is inferior to TEE” in Pag 7 line 15.

We added some information about the data regarding follow up with an appropriate statistical analysis.

We reorganized the discussion part and made the content more simple.

We added a predominantly figure regarding TTE imaging as reviewer requirement.

Thank you for your attention and help. Merry chrismas and happy new year.

Your sincerely,

Gui-chan, M.D.

2013-3-2